**Author's response to reviews**

**Title:** Income level and regional policies, underlying factors associated with unwarranted variations in conservative breast cancer surgery in Spain

**Authors:**

Manuel Ridao-López (mridao.iacs@aragon.es)
Sandra García-Armesto (sgarciaarm.iacs@aragon.es)
Begoña Abadía-Taira (mbabadia.iacs@aragon.es)
Salvador Peiró-Moreno (peiro_bor@gva.es)
Enrique Bernal-Delgado (ebernal.iacs@aragon.es)

**Version:** 7 **Date:** 14 April 2011

**Author's response to reviews:**

Dear editor,

Reaching your requirement the name of Dr Begoña Abadía-Taira is now consistent in both the electronic form and in the manuscript. In fact, the manuscript contained the proper name, being the electronic format mistaken.

My best wishes

Enrique Bernal-Delgado